Author's response to reviews

Title: The RaDIANT Community Study Protocol: Community-based Participatory Research for Reducing Disparities in Access to Kidney Transplantation

Authors:

Rachel E Patzer (rpatzer@emory.edu)
Jennifer Gander (jennifer.gander@emory.edu)
Leighann Sauls (LSauls@nw6.esrd.net)
M. Ahinee Amamoo (AAamamoo@nw6.esrd.net)
Jenna Krisher (JKrisher@nw6.esrd.net)
Laura Mulloy (LMULLOY@gru.edu)
Eric Gibney (eric.gibney@piedmont.org)
Teri Browne (BROWNETM@mailbox.sc.edu)
Stephen O Pastan (Stephen.Pastan@emoryhealthcare.org)

Version: 2
Date: 4 September 2014

Author's response to reviews:

Dear Dr. Hayley Henderson and colleagues,


Our point-by-point responses to your inquiries are below:

1. Ethical and funding approval documentation. We have emailed the requested IRB documentation and proof of funding (NIH Notice of Award) to BMCSeriesEditorial@biomedcentral.com

2. Funding – we have listed our grant number from NIH (R24MD008077) in our revised cover letter and in our manuscript. While the grant proposal for the RaDIANT Study was reviewed by NIH, this specific manuscript detailing the study protocol has not undergone peer review by the funding body.

3. Study status - Data collection for RaDIANT study is ongoing and will continue through at the earliest Spring of 2015; data analysis will continue throughout 2016.

4. Related articles - No publications containing the results of the RaDIANT study have been published or submitted to any journal.
Please feel free to contact me if you have questions or need additional materials to complete this process. Thank you for taking the time to review the letter and the manuscript. We look forward to hearing from you soon.